



UNIVERSIDAD NACIONAL DE LA PLATA

DOCUMENTOS DE TRABAJO

Abortion Legalization in Uruguay: Effects on Adolescent Fertility

Wanda Cabella y Cecilia Velázquez

Documento de Trabajo Nro. 298 Junio, 2022 ISSN 1853-0168 www.cedlas.econo.unlp.edu.ar

Cita sugerida: Cabella, Wanda y C. Velázquez (2022). Abortion Legalization in Uruguay: Effects on Adolescent Fertility. Documentos de Trabajo del CEDLAS Nº 298, Junio, 2022, CEDLAS-Universidad Nacional de La Plata.

Wanda Cabella² Cecilia Velázquez³

Abstract

The Latin America and the Caribbean adolescent fertility rate is among the highest in the world: about 1.7 million children are born to teen mothers every year, and most of them are declared as being unintended pregnancies. The region also has the highest rate of unintended pregnancy of any world region, and nearly half of such pregnancies end in abortion. However, fewer than 18% of the region's women live in countries where abortion is broadly legal. This paper estimates the causal effect of abortion legalization on adolescent fertility in Uruguay, using official data on legal abortions provided after the 2012 reform. We employed a difference-in-differences strategy, classifying states by whether they are responsive or unresponsive to the reform. The results suggest that abortion reform had a negative impact on the adolescent birth rate by 2.5 to 2.8 births per thousand adolescents aged 15–19 (4% decrease from the preintervention average). Additionally, we exploited variation in reform implementation intensity through the estimation of fixed-effect linear regression models and found consistent results. Our findings are robust to controlling for a concurrent large-scale program of contraceptive implants. We conclude that legislation aimed at enhancing rights and reducing avoidable deaths and complications from unsafe abortions may also have spillover effects that help reduce adolescent fertility.

² Programa de Población, Facultad de Ciencias Sociales, Universidad de la República del Uruguay.

¹ We are grateful to María Laura Alzúa, Pablo Gluzmann, Carolina López and Alina Machado for valuable contributions. We thank the staff of the Ministry of Health -Ima León, Ana Visconti, Rafael Aguirre and Cristina Espasandín- for providing data on legal abortions. We also thank researchers at Instituto de Economía -Rodrigo Ceni, Cecilia Parada, Ivone Perazzo and Eliana Sena- for sharing contraceptive implants data. This paper includes material from Cecilia Velázquez's Ph.D. dissertation at Universidad Nacional de La Plata. The authors bear sole responsibility for the content of this paper.

³ Centre for Distributive, Labor and Social Studies (CEDLAS-UNLP, Argentina), National Council of Scientific and Technical Research (CONICET, Argentina) and Centre for Economic Research (CINVE, Uruguay). Email: <u>cvelazquez@cedlas.org</u>.

1. Introduction

Uruguay, a small country with a population of 3.4 million, stands out in Latin America for its low religiosity, high per capita income, and low level of inequality and poverty (Latinobarometro and SEDLAC).⁴ The last two decades have witnessed significant technological and policy changes regarding abortion. At the beginning of the century, the development and widespread adoption of abortive drugs changed abortion practices worldwide, replacing surgical abortions with drug-induced abortions (Zamberlin et al., 2012). Despite legal restrictions, abortions were commonly performed in Uruguay, where women have been using misoprostol for self-induced home abortions since the early 2000s (López-Gómez, 2014). Uruguay legalized abortion in late 2012, allowing pregnancy termination during the first trimester in the health system, with all associated costs covered. This reform sets Uruguay—along with Cuba, Guyana, Puerto Rico, Mexico City, Argentina and recently Colombia—among the first places in the region where abortion on request is legal (Remez et al., 2020; Singh et al., 2018).

Abortion legalization, which has as its main expected results the full realization of sexual and reproductive rights and the reduction of unsafe abortion-related morbidity and mortality,⁵ might also have unexpected positive impacts on adolescent fertility. The effect of a policy legalizing abortion on pregnancy will depend on how abortion access affects sexual activity and the use of other methods before, during, and after a sexual encounter (Bailey and Lindo, 2018). In other words, the fertility effects of expanding access to abortion are theoretically ambiguous, because they might have a positive impact on both pregnancies and terminations (Ananat et al., 2009). Ultimately, the extent to which—or the direction in which—liberalizing abortion affects fertility remains an empirical question. This paper aims to answer this question by estimating the causal effect of abortion legalization on adolescent fertility in Uruguay, using official data on legal abortions provided after the 2012 abortion reform.

Abortion reform in Uruguay was followed by a decline in births that was more substantial among adolescents. The total fertility rate declined steadily since 2016 and reached 1.4 children per woman in 2021, consolidating Uruguay as a very low-fertility

⁴ Latinobarometro and SEDLAC databases (CEDLAS and World Bank), both consulted on March 1, 2022.

⁵ According to the classification of Singh et al. (2018), induced abortions in Uruguay were medically "least safe" before the wide spread of misoprostol in the early 2000s, "less safe" before the legalization of abortion, and "safe" afterwards. However, in this paper, we refer to safe and unsafe abortions as legal and illegal procedures, respectively.

country. The adolescent fertility rate has fallen sharply by more than half, from 61‰ in 2013 to 26 births per thousand women aged 15–19 in 2021 (see Figure 1). This significant decline distinguishes Uruguay from other Latin American and Caribbean countries.



Figure 1. Evolution of the adolescent fertility rate (Uruguay, 1961–2021)

Source: Vital Statistics, Ministry of Health and the Office of National Statistics.

Figure 1 could lead us to conclude that fertility among young women has been significantly affected by the abortion reform. However, it is not simple to assess the degree to which the adolescent fertility decline is attributable to abortion legalization rather than other concurrent policies. During this time, the government also unified several uncoordinated and isolated efforts for preventing unintended pregnancy among adolescents and articulated them into a national strategy. A central component of this strategy was the launch, in 2015, of a program offering for the first time 5-year subdermal implants at no cost to women users of the public health system (López-Gómez et al., 2019). Even controlling for all relevant concurrent policies, it is difficult to ascertain whether the decline in adolescent fertility is a direct result of abortion reform. Other factors could be driving both the fertility rates and the probability of reformed abortion laws (i.e., legislative changes and their implementation may be endogenously determined).

Building on the impact evaluation literature, researchers use policy changes relating to abortion to address endogeneity issues and isolate the causal effects of abortion access on fertility and other outcomes.⁶ Their findings indicate that changes in abortion access have an impact on fertility, with more substantial effects among adolescents and poorer women; see Bailey and Lindo (2018) for the US; González et al. (2018) for Spain; Mølland (2016) for Norway; Valente (2014) for Nepal; Pop-Eleches (2010, 2006) for Romania; and Clarke and Mühlrad (2016) and Gutiérrez-Vázquez and Parrado (2016) for Mexico City.

Following its legalization of abortion, Uruguay represents an exciting case study. There is still scarce research on the effects of the abortion reform, partly because it is very recent and also because access to information is limited. Most of the reviewed papers have analyzed the progress of the law from different perspectives, particularly from health sciences (Fiol et al., 2016; Labandera et al., 2016); social sciences (Ituarte and López-Gómez, 2021; López-Gómez and Couto, 2017); and the implementation of legal abortion services (MYSU, 2014, 2015, 2016a, 2016b, 2017). To our knowledge, Antón et al. (2018) is the only previous study that has used impact evaluation techniques. They found an 8% decline in the number of births from unplanned pregnancies driven by the group of women aged 20–34 with completed secondary education, an improvement in prenatal care and a smaller share of single mothers. However, they did not find any impact among adolescents.

After this introduction, the rest of the paper is organized as follows: Section 2 provides the background for adolescent fertility trends in Uruguay and presents a description of the abortion reform. Section 3 describes the data used in this study and lays out the methodology. Section 4 presents the main findings, and section 5 concludes.

⁶ In addition to the fertility effects, changes to abortion access may affect outcomes for women and the next generation (Azevedo et al., 2012; Bernstein and Jones, 2019). First, empirical evidence about the effect for women facing different abortion policies on their educational and labor prospects is significant and positive in Norway (Mølland, 2016) and Spain (González et al., 2018). In the United States, the evidence is mixed, with some studies finding no effect (Angrist and Evans, 1999; Ribar, 1994), while others show the strong long-lasting negative impact of denying abortion to adolescents. This is, for example, the conclusion of the large longitudinal study Turnaway, the main result of which is that denying women abortions leads to poorer economic, health, and family outcomes (Foster, 2019). Second, the empirical evidence shows significant effects for the cohort of affected children in the US (Ananat et al., 2009; Donohue and Levitt, 2001; Gruber et al., 1999); Norway (Mølland, 2016); and Romania (Pop-Eleches, 2006). However, no intergenerational effects were found in a high-fertility country such as Nepal (Valente, 2014).

2. Adolescent fertility and abortion in Uruguay

2.1. Evolution of adolescent fertility

The persistence of high adolescent fertility rates is a peculiar characteristic of the Latin American fertility regime, although the Total Fertility Rate (TFR) has been declining steadily since the 1970s (Cabella and Pardo, 2014; ECLAC, 2011). The adolescent fertility rate in the region is one of the highest in the world, second only to that of the African continent. Latin America also has the highest rate of unintended pregnancy in any world region, and more than 24 million women face an unmet need for modern contraception, as indicated in the work of Singh et al. (2018). The same report estimates that 14 million unintended pregnancies occur each year, 46% of which end in abortion. However, fewer than 18% of the region's women live in countries where abortion is broadly legal.⁷ The unwanted adolescent fertility rate on the continent is also the highest globally, reaching 50%, and has increased in recent years (Rodríguez-Vignoli, 2017).

Uruguay faced a high adolescent fertility rate between the 1990s and the mid-2010s, well above the level expected considering its low total fertility rate and its socioeconomic indicators (Rodríguez-Vignoli and Cavenaghi, 2014; Varela-Petito et al., 2014). This trend reversed in 2016 when the country witnessed a steep reduction in adolescent and very young fertility. After more than 40 years of cycles of increase–decrease–stagnation at high levels, adolescent fertility is now leading the decline between 2015 and 2018: 29% of the fall in TFR is explained by the reduction in adolescent fertility, and the fall in the rate between the ages of 15 and 22 years explains 50% of the decline in the TFR, which in 2018 reached 1.59 children per woman (Cabella et al., 2019).

The stagnation of adolescent fertility is one of the aspects of Uruguayan sexual and reproductive health that has received the most attention during the last 15 years, both from academia and in the field of public health, particularly because it reflects strong social polarization in reproductive behavior (Cardozo and Iervolino 2009; Nathan et al., 2016; Varela-Petito et al., 2012, 2014; Videgain, 2006). Adolescent girls from the

⁷ Remez et al. (2020) and Singh et al. (2018) report that fewer than 3% of women of reproductive age in Latin America and the Caribbean lived in places where abortion was broadly legal, that is, permitted either without restriction due to reason (Cuba, Guyana, Puerto Rico, Mexico City and Uruguay) or on socioeconomic grounds (Barbados, Belize and St. Vincent and Grenadines). This share was updated to 18% using female population 15–49 data by age group from United Nations (2019) to include Argentinian and Colombian recent reforms.

higher social sectors tend to start their sexual life later (Cabella et al., 2017), rely more frequently on contraception, and use it consistently; thus, they have their children later and face fewer births that are earlier than desired. Adolescent motherhood is a phenomenon that is almost exclusive to poor settings and is generally associated with inconsistent use of contraceptive methods and reluctance to carry out an abortion (Brunet et al., 2019); according to official data, 70% of adolescent mothers report that their pregnancies were unwanted or mistimed (Cabella et al., 2017).

Differences in the timing of first birth are strongly rooted and have deepened during the process of fertility decline. These differences are anchored in educational and socioeconomic inequalities that define the centrality of motherhood in young women's life alternatives (Nathan, 2015; Varela-Petito et al., 2012). Adolescent girls with a low level of education, who are mostly outside the education system, perceive the arrival of a child as a factor of entrenchment of their social identity and a way to reduce the uncertainty and lack of security provided by the social environment (Amorin et al., 2006; De Rosa et al., 2016). Women with a higher level of education face a set of alternatives that are more attractive than motherhood at that stage of life, while, for less-educated women, there are fewer incentives to give up the reproductive project, which in many cases is the only one they envision.⁸

Adolescent fertility also shows important territorial variations.⁹ As shown in Figure 2, it is higher in the northern and northeastern states of the country (Artigas, Salto, Paysandú, Rivera, Río Negro, and Tacuarembó) where economic and social development is lower. However, it is worth noting that the recent drop in adolescent fertility rate occurred in all states. At different rates and starting from different levels, it is evident that they are all converging toward a lower adolescent fertility rate.

⁸ Human capital accumulation and reproductive decisions are either joint decisions, which result in a potential reverse causality problem or are both affected by unobservable factors, causing selection bias. Alzúa and Velázquez (2017), Berthelon and Kruger (2011), Cortés et al. (2016, 2010) and Novella and Ripani (2016) provide causal evidence for Argentina, Chile, Colombia and the Dominican Republic that supports the hypothesis that education reduces fertility among youths.

⁹ Uruguay is organized as a unitary state divided into 19 political units called departments. They differ widely in population size; 60% of the population is concentrated in the capital, Montevideo, and its metropolitan area, while the departments in the center of the country are virtually empty (Durazno, Flores, Lavalleja).



Figure 2. Evolution of the adolescent fertility rate by state (Uruguay, 2006–2018)

Source: Vital Statistics, Ministry of Health and the Office of National Statistics.

It should be noted that the strong annual variations observed in some states are related to their small population.

2.2. Legal framework and implementation of abortion reform

Abortion in Uruguay was briefly decriminalized between 1934 and 1938. Since then, it has been punishable by law and only permitted to save the woman's life and to preserve her health. Although the law mentioned rape, "family honor," and "economic problems" as grounds for abortion, exemptions were very rare, and services that would provide abortions under the established grounds were never implemented. Since 1985, the topic entered the public agenda, and all subsequent parliaments have considered bills on abortion. Even in 2008, a bill legalizing abortion was passed by parliament but promptly vetoed by the president in a highly contested decision (López-Gómez and Abracinskas, 2009).

Despite legal restrictions, since 2001 health professionals of the main public maternity hospital in Montevideo began to provide counselling and care to women who wanted to terminate a pregnancy, both before and after a clandestine abortion. These professionals informed women about the use of abortive drugs (mainly misoprostol), but they were legally restricted from providing them (Fiol et al., 2012; Labandera et al.,

2016). Misoprostol could be purchased on the black market since 2002, although at very expensive prices. Since its availability, surgical abortions at clandestine clinics have been replaced by drug-induced abortions at home (López-Gómez, 2014).

Uruguay passed its abortion law reform in December 2012 (Law 18987 of Voluntary Termination of Pregnancy). The reform allows for pregnancy termination in the health system with all associated costs covered, upon request until the 12th week of pregnancy, in the cases of rape until the 14th week, and at any time during pregnancy in cases of risk of maternal death or fetal anomalies incompatible with life.

Three consultations are required before the termination of unwanted pregnancies. During the first consultation with the gynecologist, the professional must verify that the gestational age requirements are met. The gynecologist schedules the second consultation within the next 24 hours, with a board composed of a gynecologist, a social worker, and a psychologist who provide women with information on the decision (risks of the procedure, alternative options, and social support programs for maternity or adoption). After a mandatory waiting period of 5 days, the woman can confirm her decision, and the gynecologist determines the most suitable procedure. The vast majority (98.8%) of legal abortions were medication-induced abortions (mifepristone and misoprostol regimen), and almost 97% of those women used the medication as outpatients (Fiol et al., 2016). A fourth and final consultation for a medical check and for providing a contraceptive method is scheduled 10 days after the abortion.¹⁰

The number of abortions registered in health services increased by 45% in the 6 years between 2013 and 2018 (Table 1). The legal abortion rate (the number of legal abortions per thousand women aged 15–49) increased from 8.6 to 12.1 in the same period. However, the growth rate has slowed down, as it has in other countries where abortion has been decriminalized (see for Mexico Darney et al., 2019; and for a more general reference Singh et al., 2018). The overall annual increase in legal abortions among adolescents aged 15–19 was significantly lower, registering an 18% increase between 2013 and 2018. The legal abortion rate in this age group increased from 8.9 in 2013 to 11.2 in 2018. Following the rise during the first 2 years of abortion reform implementation, the annual number of legal abortions has tended to stabilize or even decline within this age group.

¹⁰ Adolescents below the legal age of majority (18 years old) can decide to terminate a pregnancy if the threeprofessional board approves it after assessing her progressive autonomy.

	Total: women aged 15-49				Teenagers: women aged 15-19			
Year	Legal Abortions	Live Births	Legal Abortion Rate	Total Fertility Rate	Legal Abortions	Live Births	Legal Abortion Rate	Adolescent Fertility Rate
2013	7171	48681	8.6	1.96	1200	8172	8.9	60.9
2014	8537	48368	10.1	1.94	1404	7779	10.5	58.1
2015	9362	48926	11.1	1.96	1603	7371	12.1	55.6
2016	9719	47058	11.4	1.88	1597	6575	12.2	50.3
2017	9830	43036	11.5	1.71	1476	5367	11.5	41.6
2018	10373	40139	12.1	1.60	1421	4554	11.2	35.8

 Table 1. Legal abortions and births (Uruguay, 2013–2018)

Notes: The legal abortion rate represents the total number of legal abortions per 1,000 women aged 15–49 in the left panel and the number of legal abortions among adolescents per 1,000 women aged 15–19 in the right panel. Source: Based on information about legal abortions provided by the Ministry of Health, Vital Statistics (Ministry of Health and Office of National Statistics of Uruguay), and mid-year population estimates (Office of National Statistics of Uruguay).

Although abortion is legal nationwide following the reform, there are still barriers to access. The law requires abortions to be provided in health services by gynecologists, but allows them to refuse due to issues of conscientious objection. Regulatory changes regarding conscientious objection, combined with the fact that many regions of the country face a shortage of physicians specializing in gynecology,¹¹ result in considerable variation across states over time in the actual provision of abortion services (López-Gómez and Couto, 2017). A particular example is the case of the state of Salto, where the 12 resident gynecologists invoked conscientious objection, leaving women without access to abortion services for several months (MYSU, 2014).

Unfortunately, there is no complete information at the state level about the evolution of conscientious objectors. The scarce official information shows that, in the first year after the reform (2013), 30% of Uruguayan gynecologists declared themselves to be objectors. This rose to 40% in 2015 (data provided by the Ministry of Health).

In those cities or towns where all gynecologists refuse to provide abortions due to conscientious objection, women must be referred to other cities in the country for care. In all cases, when services are overcrowded, women are referred to the main public maternity hospital in Montevideo. The travel costs must be covered by the institution that fails to provide the service, although this is not always the case (MYSU, 2014,

¹¹ According to the latest information provided by the Ministry of Health, Uruguay has a total of about 650 gynecologists. This yields a rate of approximately 0.4 gynecologists per thousand women aged 14 or above. The physicians specializing in gynecology are unevenly distributed across the country and are highly concentrated in the country's capital Montevideo.

2015, 2016a, 2016b, 2017). The appendix presents further details about the evolution of regulations and the geographical distribution of conscientious objection.

Despite the difficulties in implementing the law, it is important to note that maternal mortality has improved since the abortion law was passed.¹² From legalization (2012) until 2019, there have been a total of 57 maternal deaths in the country, of which 4 were in adolescents. In the last four years (2016–2019), there were no maternal deaths among adolescents according to official data (Ministry of Health).

¹² Uruguay has stood out in the region for its very low level of maternal mortality for several decades. In the recent period, this has continued to decrease, and, since the law was passed, it has experienced a further decline.

3. Empirical strategy

3.1. Data

Data about legal abortions in Uruguay are not publicly available. For this paper, we have accessed the information on abortions for the period from December 2012 to December 2018 through an Agreement with the Ministry of Health.¹³ The Ministry of Health has gathered data on legal abortions from each private and public health center monthly since December 2012. Health centers must report the total number of women who attend the second, third, and fourth consultations by aggregate age group (14 or younger, 15–19, 20–44, and 45 and over). Unfortunately, abortion information is collected at the health center level and there is no information at the individual level, nor is there any aggregate information about women, such as previous births or abortions, contraceptive use, marital status, or place of residence. To merge abortion data with information on births, we aggregated the data at the state level based on each health center's location. Due to the low number of observations, we aggregated monthly data on a quarterly basis.

We thus constructed a balanced panel of quarter state-level data from 2006 to 2018. The total number of quarter-state observations is 988 (19 states, 4 observations each year, 13 years). For the period 2006 to the third quarter of 2012 (i.e., pre-reform), the number of legal abortions (and the legal abortion rate) is zero. We discuss the implications of omitting clandestine abortions in the Discussion (section 4.3).

To examine the effects of the abortion reform on fertility, we used the Vital Statistics on all births registered in Uruguay. Birth data are compiled by the Ministry of Health and the Office of National Statistics based on live-birth certificates issued at hospitals. We focused on the number of live births by the age and state of residence of the mother.

We calculated the adolescent legal abortion rate as the number of legal abortions among adolescents aged 15–19 per thousand women aged 15–19 years. Likewise, we computed the adolescent fertility rate as the number of live births among adolescents aged 15–19 per thousand women aged 15–19 years. The population of women (variation by age group, year, and state of residence) was taken from mid-year

¹³ Under this agreement, we did not have access to women's identified information, nor the possibility of linking it to their medical records. For this reason, the Ministry of Health did not consider necessary the approval of an ethics committee to provide us with the data.

population estimates (Office of National Statistics of Uruguay, 2013 Revision). Quarterly rates were annualized by multiplying by 4.

An important concern regarding attempts to establish causality is the existence of interventions occurring at the same time as the abortion reform, such as the contraceptive implant program launched in 2015. To deal with this issue, we used two different measures for the subdermal implant program adoption. First, we employed a dummy variable taken from Ceni et al. (2021), indicating the dates of the program's start in each state. Second, we included a measure based on information about subdermal implant shipments from the Ministry of Health to health facilities in all states. This information is provided by Ceni et al. (2021); according to their estimations 80% of women using the implant are under 30 years of age. We used the accumulated stock of implants shipped as a proxy variable for women using implants,¹⁴ and we calculated the rate taking women aged 15–29 as the reference population.

Bearing in mind that abortion is allowed until the 12th week of pregnancy, the abortion law rules out those women who discover that they are pregnant after this limit. The gestational age at the beginning of prenatal care among live births to adolescent mothers, available in the Vital Statistics, serves as a proxy for the time it usually takes for a young woman to realize that she is pregnant.

Finally, we merged a set of time- and state-varying covariates for each quarter and state. This includes the youth unemployment rate (15–24 years old)—from Uruguay's national household survey conducted by the Office of National Statistics—as an economic cycle indicator. We also used the number of public secondary schools per 10,000 inhabitants, from the Education Statistics National Yearbooks (Ministry of Education) as a proxy for the state schooling infrastructure. The number of physicians and the number of nurses per 10,000 inhabitants, from the records of the Pension and Retirement Fund of University Professionals, were also included as a proxy for state health resources. Finally, we used the state share of the total population from mid-year population estimates (Office of National Statistics of Uruguay, 2013 Revision) as an indicator of the state's size. It should be noted that the number of schools, physicians, nurses, and the population varies yearly, not quarterly.

¹⁴ The accumulated stock of implants shipped can serve as a proxy variable for women using implants under the following assumptions: (a) no new implants are sent until the stocks are exhausted; (b) women do not travel out of state to access implants; and (c) all implants shipped are used and there are no removals (Ceni et al., 2021). The available data from the pilot program carried out in 2014 show high continuation rates (fewer than 10% of the women removed the implant after one year) that have been rising thereafter (López-Gómez et al., 2019).

3.2. Estimation strategy

Since the abortion reform in Uruguay was not accompanied by any strategy to evaluate its impact causally, we had no other choice but to rely on observational data to derive our results. First, we employed a difference-in-differences framework to determine whether states more responsive to the reform performed better in reducing adolescent fertility. Second, we exploited variation in treatment intensity through the estimation of fixed-effect linear regression models, regressing adolescent fertility rates on lagged legal abortion rates for the same age group. Third, we implemented a robustness test to address concerns about regional mobility. All specifications control for state fixed effects, time dummies, and state-average socioeconomic and demographic variables, as well as including two alternative measures of contraceptive implants.

We use a difference-in-differences strategy because time variation in the application of the law across states implies that women of the same age but living in different states had different levels of access. Some women would have faced more significant barriers to obtaining legal abortion services according to the time and their place of residence, and this could have a differential effect between states in fertility rates.

Based on the average of post-reform legal abortion rates from adult women aged 20– 44, we split states in treatment and control using the median to obtain two groups of similar size. We referred to the states above the median as "responsive" and the other states as "unresponsive" to the reform—that is, treatment and control group, respectively. The states in each group can be observed in Figure 3. We then compared the adolescent fertility rates of responsive states with unresponsive states, before and after the reform implementation.

We assume that the sub-national variation in legal abortion rates of adult (20–44) women is driven by idiosyncratic variation in the provision of abortion services at the state level, related to the regulation and use of conscientious objection (i.e., supply factors) and unrelated to adolescent's decisions (i.e., demand factors). This identifying assumption is quite reasonable, because supply-side constraints are fundamental, particularly during the reform's initial months. The decision to invoke conscientious objection responds to economic and corporate interests, led by some gynecologists who are opponents to the reform (López-Gómez and Couto, 2017), and is therefore

13

exogenous to adolescent's reproductive decisions. See the appendix for further details regarding context on conscientious objection.

We used the balanced-panel quarter state-level dataset described in the previous section and estimated the following equation:

(1) Fertility_{it} = $\alpha_1 Responsive_i * Post_t + \alpha_2 \rho_i + \alpha_3 \sigma_t + \alpha_4 X_{it} + \omega_{it}$

where *Fertility_{jt}* is the number of live births among adolescents during quarter *t* multiplied by 4 per thousand women aged 15–19 years in state *j* (i.e., the annualized adolescent fertility rate). *Responsive_j* is a binary variable taking the value of 1 if the state is responsive to the abortion law reform and 0 otherwise, and *Post_t* is a dummy equal to 1 in the post-reform period (second quarter 2013 onwards) and 0 otherwise (although the law was passed in December 2012, we must bear in mind that current abortions impact births 5–7 months later). *Responsive_j*Post_t* is an interaction term between these variables.

Evaluation of the impact of a new law involves several challenges. Of particular concern is the fact that the implementation of the abortion reform may be endogenously determined. That is, the sub-national legal abortion rates are likely to be correlated with observed and unobserved state characteristics. The inclusion of state fixed effects (ρ_j) removes any time-invariant heterogeneity, because all observable and unobservable state-specific invariant factors are absorbed into the fixed effect. We also included time dummies (σ_t) to reduce the correlation across states and control for common factors that affect fertility across all states within a specified quarter. We also introduced a set of time-varying controls at the state level (X_{jt}), detailed in the previous section. To address potential heteroskedasticity, serial correlation, and cross-sectional correlation issues in the idiosyncratic error term (ω_{jt}), we estimated Driscoll and Kraay's standard errors.¹⁵

¹⁵ We did not perform estimations with clustered standard errors at the state level as is customary because few clusters mean biased standard errors and misleading inferences; that is, 19 groups are not enough for reliable inference (Angrist and Pischke, 2008). Driscoll and Kraay (1998) propose a nonparametric covariance matrix estimator that produces heteroskedasticity- and autocorrelation-consistent standard errors that are robust to general forms of spatial and temporal dependence. We implemented Driscoll and Kraay's standard errors using the Stata XTSCC module (Hoechle, 2007). The maximum lag order of autocorrelation was set at 3, determined by Hoechle's (2007) rule of thumb: $floor[4(T/100)^{2/9}]$.

An important concern regarding the identification strategy is the existence of interventions occurring at the same time as the abortion reform, specifically the contraceptive implant program launched by the Uruguayan Government in 2015.¹⁶ To deal with this issue, we sought additional information about this program. We estimated the following equations:

(2) Fertility_{it} = $\beta_1 Responsive_i * Post_t + \beta_2 ImplantsDummy_{it-k} + \beta_3 \rho_i + \beta_4 \sigma_t + \beta_5 X_{it} + \omega_{it}$

(3) Fertility_{it} = $\gamma_1 Responsive_i * Post_t + \gamma_2 ImplantsUse_{it-k} + \gamma_3 \rho_i + \gamma_4 \sigma_t + \gamma_5 X_{it} + \omega_{it}$

As discussed above, we used two types of variables for implants: *ImplantsDummy*_{*jt-k*}, a binary variable indicating the start of the subdermal contraceptive program in each state and *ImplantsUse*_{*jt-k*}, the share of women aged 15–29 using implants. Given that the current insertion of contraceptive implants will prevent births 9 months later and thereafter, we included implant variables lagged by 9 months (i.e., *k=9 months*). It is worth pointing out that we lagged the monthly data and then aggregated them quarterly.

The above difference-in-differences model assumes adolescent fertility trends (conditional on *X*) would have been the same in both groups of states in the absence of the abortion reform. This assumption is necessary to obtain a causal interpretation of the estimates of α_1 , β_1 , and γ_1 in equations (1) to (3). The parallel or common trends assumption is inherently non-testable, but a good validity check is to compare the trends in pre-treatment periods (see Figure 4 in section 4.1.).

Second, to provide additional evidence of our results, we estimated fixed-effect linear regression models, regressing adolescent fertility rates on lagged legal abortion rates for the same age group. Up to this point, we compared two groups of states with high and low legal abortion rates for women aged 20–44, we then exploited variation in legal abortion rates for adolescents (variation in treatment intensity). We estimated the following equation:

(4) *Fertility*_{*jt*} = $\delta_1 Abortion_{jt-l} + \delta_2 \rho_j + \delta_3 \sigma_t + \delta_4 X_{jt} + \omega_{jt}$

¹⁶ Another relevant intervention is the program *Metas asistenciales* (which uses a pay-for-performance scheme). This program has fostered the early detection of pregnancy since 2008, and as a result prenatal visits have become increasingly early. We included in our estimates the gestational age at the beginning of prenatal care among live births to teen mothers.

where *Abortion*_{*jt-1*} is the number of legal abortions among adolescents during quarter *t*– *l* multiplied by 4 per thousand women aged 15–19 years in state *j* (i.e., the lagged value of the annualized adolescent legal abortion rate). Bearing in mind that termination is allowed until the 12th week of pregnancy while the length of adolescent pregnancy is about 38 weeks, current abortions will affect fertility rates approximately 6 months later. We used the adolescent legal abortion rate lagged by 5, 6, and 7 months (i.e., we have different specifications for *I*=5, *I*=6, and *I*=7 months). We lagged the monthly data and then we aggregated them quarterly, as we did with implant contraceptive variables.

All specifications control for state fixed effects (ρ_j), time dummies (σ_t), and the same set of state-average socioeconomic and demographic variables listed above (X_{jt}). Again, we included two alternative measures of contraceptive implants:

(5) Fertility_{it} =
$$\tau_1 Abortion_{it-1} + \tau_2 ImplantsDummy_{it-k} + \tau_3\rho_i + \tau_4\sigma_t + \tau_5X_{it} + \omega_{it}$$

(6) Fertility_{it} =
$$\pi_1 Abortion_{it-l} + \pi_2 ImplantsUse_{it-k} + \pi_3\rho_i + \pi_4\sigma_t + \pi_5X_{it} + \omega_{it}$$

Third, we implemented a robustness test to address concerns about regional mobility. Our measure of abortion is based on the health facility location where the woman attends the third consultation to terminate her pregnancy, but we do not have information about her state of residence. As discussed above, in those cities where the share of objectors among gynecologists is high, women must be referred to other cities in the country for attention. Some women may also prefer to travel out of town, seeking anonymity, given the social stigma that still surrounds abortion. To address this concern about regional mobility, we performed a robustness test estimating the following equations:

(4') Fertility_{it} =
$$\delta'_1$$
 Pre-AbortionVisit_{it-l} + $\delta'_2 \rho_i$ + $\delta'_3 \sigma_t$ + $\delta'_4 X_{it}$ + ω_{it}

(5') Fertility_{it} =
$$\tau'_1$$
 Pre-AbortionVisit_{it-l} + τ'_2 ImplantsDummy_{it-k} + $\tau'_3\rho_i$ + $\tau'_4\sigma_t$ + τ'_5X_{it} + ω_{it}

(6') Fertility_{jt} =
$$\pi'_1$$
 Pre-AbortionVisit_{jt-l} + π'_2 ImplantsUse_{jt-k} + $\pi'_3\rho_j$ + $\pi'_4\sigma_t$ + π'_5X_{jt} + ω_{jt}

where *Pre-AbortionVisit*_{*jt-1*} is the number of pre-abortion consultations during quarter *t- I* multiplied by 4 per thousand women aged 15–19 in state *j* (i.e., the lagged value of the annualized adolescent pre-abortion rate). Note that, in this case, state *j* is based on the health facility location where the woman attends the second consultation—that is, the counseling consultation (while *Abortion*_{*jt-1*} is based on the location where the woman attends the third consultation). Because there is a mandatory five-day waiting period between the second and the third consultation, it is more likely that a woman will attend the second consultation in her place of residence and then travel to another city to attend the third consultation.

4. Results

4.1. Difference-in-differences estimates

The first step for employing the difference-in-differences method is to classify the states into two groups. Figure 3 distinguishes responsive (treatment) and unresponsive (control) states.



Figure 3. Treatment and control groups of states

Table 2 shows the means of adolescent fertility rates and those of the state socioeconomic and demographic variables for each group of states. It also gives the difference between the two groups in the pre-reform period. As can be seen, states differ significantly in many dimensions. In general, adolescent fertility rates, the number of physicians per inhabitant, and the youth unemployment rate are higher for the states in the treatment group, which also have a larger population. Treatment states also present a lower number of secondary schools per inhabitant, and women attend prenatal visits later, on average, than the control group. These pre-existing differences indicate the importance of controlling for these state characteristics for unbiased estimations.

Notes: we used the median to obtain two groups of similar size –based on the average of post-reform legal abortion rates from women aged 20-44- and classified states by whether they are responsive or unresponsive to the abortion law reform.

Variables	Treat (i)	Control (ii)	Diff. (ii)-(i)	t	p-value
Teenage (15-19) fertility rate (‰), annualized	70.66	61.30	-9.36	-8.09	0.00
Number of physicians per 10,000 inhabitants	24.49	21.60	-2.89	-2.62	0.01
Number of nurses per 10,000 inhabitants	8.15	8.20	0.05	0.13	0.89
Number of public secondary schools per 10,000 inhabitants	1.23	1.33	0.10	3.10	0.00
Unemployment rate (%), 15-24 years old	22.35	21.19	-1.16	-1.72	0.09
State share of total population (%)	6.41	3.98	-2.43	-3.25	0.00
Beginning prenatal care among live births to teen mothers (average weeks pregnant)	15.23	14.54	-0.69	-5.13	0.00
Observations	290	261			•

Table 2. Differences in means, Treatment and control group. Pre-reform period.

Notes: The table reports simple state-level averages for quarters in the sample before the reform (2006 to 1st quarter 2013) for treatment (column 1) and control group (column 2), and tests on the equality of means between both groups (columns 3 to 5).

Figure 4 presents the adolescent fertility rates for responsive and unresponsive states conditional on state-average socioeconomic and demographic variables and contraceptive implants. Adolescent fertility levels are lower for the unresponsive group of states than for the responsive states. Despite the difference in levels, both groups exhibit similar trends during the pre-reform period, and there is a reduction in the gap between the two groups after the law was passed.





Notes: The graph shows the predicted means trends for adolescent fertility rates for responsive (treatment group) and unresponsive states (control group) using OLS regressions by year conditional on state-average socioeconomic and demographic controls and a measure of contraceptive implants. Covariates: a dummy variable indicating for each state the implants program start, the youth unemployment rate, number of secondary schools per 10,000 inhabitants, number of physicians and nurses per 10,000 inhabitants, state share of the total population, and average weeks pregnant when beginning prenatal care among live births to teen mothers.

Data presented in Figure 4 suggest that those states responsive to the abortion law reform performed better in reducing adolescent fertility. The econometric estimates of this effect are reported in Table 3: the first column displays coefficient α_1 from equation (1); column 2 presents coefficients β_1 and β_2 from equation (2); and column 3 presents γ_1 and γ_2 from equation (3). The last row in Table 3 reports the simple state-level average of adolescent fertility rates for the period before the implementation of the abortion reform (2006 to first quarter 2013).

The treatment effect in column 1 represents the effect of abortion legalization on fertility among teenagers conditional on state fixed effects, time dummies, and the set of state-specific time-varying controls listed above. The result indicates a statistically significant decrease in adolescent fertility rates for the group in states responsive to the abortion reform relative to the unresponsive group of around -2.8 births per thousand women aged 15–19.

Columns 2 and 3 also account for two alternative measures for the contraceptive implant program 9 months earlier. We observe that accounting for this program produces a qualitatively similar set of results. The coefficients for abortion reform remain negative, statistically significant, and quite similar to those reported in column 1, although smaller. To sum up, the abortion reform has had a negative impact on adolescent births rates by 2.5 to 2.8 births per thousand adolescents aged 15–19. In terms of the pre-intervention average, this effect represents a 4% decrease in the adolescent fertility rate.

We also considered the coefficients for contraceptive implant measures. In column 2, the dummy variable indicating the program launch for each state, significant at the 10% level, shows a reduction of 1.5 births per thousand women aged 15–19. In column 3, we include the share of women aged 15–29 using implants instead of the dummy variable, the coefficient results significant at the 1% level, implying that an increase of one percentage point in the implant use rate reduces by about 0.13 the number of births per thousand girls (aged 15–19) 9 months later.

	(1)	(2)	(3)
Responsive x Post	-2.796*	-2.654*	-2.495*
	(1.441)	(1.428)	(1.404)
Subdermal Implants (t-9m)		-1.544*	-0.126***
		(0.865)	(0.030)
Dependent variable mean	66.2	66.2	66.2
Observations	988	988	988
State Fixed Effects	Yes	Yes	Yes
Time Dummies	Yes	Yes	Yes
Covariates	Yes	Yes	Yes
Subdermal Implants (t-9m)		Program start (dummy)	Women 15-29 using implants (%)

Table 3. Effect of abortion legalization and the contraceptive implant program on fertility rates among women aged 15–19

Notes: The results are from estimating equations (1) to (3) in columns 1 to 3 respectively. The table displays the coefficients of the interaction term $Responsive_i *Post_i$. In columns 2 and 3, the table also displays the coefficients of the variables for $Implants_{it-k}$ lagged by 9 months: a dummy variable indicating for each state the program start (column 2) and a proxy for the percentage of women aged 15–29 using implants (column 3). Covariates: the youth unemployment rate, number of secondary schools per 10,000 inhabitants, number of physicians and nurses per 10,000 inhabitants, state share of the total population, and average weeks pregnant when beginning prenatal care among live births to teen mothers. The reported mean for the dependent variable corresponds to the pre-reform period (2006 to 1st quarter 2013). Driscoll–Kraay standard errors in parentheses. * p<0.10, ** p<0.05, *** p<0.01.

4.2. Regression of fertility rates on lagged legal abortion rates for teenagers

Table 4 reports the estimations of equations (4) to (6). The first column displays coefficient δ_1 in equation (4); column 2 presents coefficients τ_1 and τ_2 in equation (5); and column 3 presents π_1 and π_2 from equation (6). Each panel shows the estimated coefficients of the variable *Abortion*_{*j*t-*l*} with *l*=5, *l*=6, and *l*=7 months in panels A, B, and C, respectively.

Column 1 represents the effect of a one-point increase in the legal abortion rate on fertility among adolescents, controlling for the state fixed effect, time dummies, and the set of covariates listed before. We find that a one-point increase in the adolescent legal abortion rate reduces—5, 6, and 7 months later—the number of births per thousand girls aged 15–19 by about 0.18 to 0.24.

Columns 2 and 3 also account for two alternative measures for the contraceptive implant program 9 months earlier. Again we observe that accounting for this program

produces a qualitatively similar set of results. The coefficients for abortion remain negative, statistically significant, and roughly equal to those reported in column 1.

We also considered the coefficients for implants contraceptive measures. In column 2, the dummy variable indicating the program launch for each state, significant at the 10% level (except for panel C, which reports a statistically insignificant coefficient for implants), ranges from -1.7 to -1.6. In column 3, we include the share of women aged 15–29 using implants instead of the dummy variable, and the coefficients are significant at the 1% level, implying that an increase of one percentage point in the implant use rate reduces 9 months later the number of births per thousand girls aged 15–19 by about 0.13.

	(1)	(2)	(3)
Panel A			
Abortion Rate (t-5m)	-0.239***	-0.234***	-0.238***
	(0.083)	(0.083)	(0.083)
Subdermal Implants (t-9m)		-1.704*	-0.133***
		(0.948)	(0.029)
Panel B			
Abortion Rate (t-6m)	-0.178*	-0.171*	-0.174*
	(0.090)	(0.090)	(0.089)
Subdermal Implants (t-9m)		-1.669*	-0.131***
		(0.957)	(0.029)
Panel C			
Abortion Rate (t-7m)	-0.198**	-0.190*	-0.192**
	(0.098)	(0.098)	(0.095)
Subdermal Implants (t-9m)		-1.589	-0.131***
		(0.993)	(0.029)
Observations	988	988	988
State Fixed Effects	Yes	Yes	Yes
Time Dummies	Yes	Yes	Yes
Covariates	Yes	Yes	Yes
Subdermal Implants (t-9m)		Programme start (dummy)	Women 15-29 using implants (%)

 Table 4. Effects of abortion legalization and the contraceptive implant program on fertility rates among women aged 15–19

Notes: The results are from estimating equations (4) to (6) in columns 1 to 3 respectively. The table displays the coefficients of the variable *Abortion_{jt-1}* (the lagged value of the annualized adolescent legal abortion rate) for *I=5*, 6 and 7 months in panels A, B and C, respectively. In columns 2–3, the table also displays the coefficients of the variables for *Implants_{jt-k}* lagged by 9 months: a dummy variable indicating for each state the program start (column 2) and a proxy for the percentage of women aged 15–29 using implants (column 3). Covariates: the youth unemployment rate, number of secondary schools per 10,000 inhabitants, number of physicians and nurses per 10,000 inhabitants, state share of the total population, and average weeks pregnant when beginning prenatal care among live births to teen mothers. Driscoll–Kraay standard errors in parentheses. * p<0.10, ** p<0.05, *** p<0.01.

As discussed in section 3.2, we used the state legal abortion rates of adolescents, which are based on the health facility location where the women attend the third consultation to terminate their pregnancy, but we do not have information about their state of residence. We re-estimate equations (4) to (6) using the number of pre-abortion consultations (i.e., the second counseling consultation with the three-professional board) per thousand women aged 15–19 as a proxy for women's place of residence. Table 5 reports the estimations of equations (4') to (6').

	(1)	(2)	(3)
Panel A			
Pre-abortion visit rate (t-5m)	-0.182*	-0.177*	-0.180*
	(0.102)	(0.101)	(0.101)
Subdermal Implants (t-9m)		-1.727*	-0.132***
		(0.962)	(0.029)
Panel B			
Pre-abortion visit rate (t-6m)	-0.122	-0.115	-0.117
	(0.094)	(0.093)	(0.093)
Subdermal Implants (t-9m)		-1.750*	-0.132***
		(0.947)	(0.029)
Panel C			
Pre-abortion visit rate (t-7m)	-0.140	-0.132	-0.133
	(0.096)	(0.095)	(0.094)
Subdermal Implants (t-9m)		-1.671*	-0.131***
		(0.990)	(0.028)
Observations	988	988	988
State Fixed Effects	Yes	Yes	Yes
Time Dummies	Yes	Yes	Yes
Covariates	Yes	Yes	Yes
Subdermal Implants (t-9m)		Programme start (dummy)	Women 15-29 using implants (%)

Table 5. Robustness check

Notes: The table replicates the results from Table 4, replacing the variable Abortion_{*j*t-*l*} (the lagged value of the annualized adolescent legal abortion rate) with *Pre-AbortionVisit_jt-l* (the lagged value of the annualized number of pre-abortion counseling consultations per 1,000 women aged 15–19). Covariates: the youth unemployment rate, number of secondary schools per 10,000 inhabitants, number of physicians and nurses per 10,000 inhabitants, state share of the total population, and average weeks pregnant when beginning prenatal care among live births to teen mothers. Driscoll–Kraay standard errors in parentheses. * p<0.10, ** p<0.05, *** p<0.01.

The results in Table 5 are similar to those in Table 4, although the coefficients are less significant and smaller. However, it is worth mentioning that these differences are not statistically significant. Figure 5 shows the estimated coefficients accompanied by a 90% confidence interval. The overlapping of confidence intervals in all cases indicates that the coefficients for the legal abortion rates (Table 4) are not statistically different from the coefficients for the pre-abortion visits for mandatory counseling prior to

abortion (Table 5). This addresses the concern that regional mobility across states might bias our main results.



Figure 5. Comparison of estimated effects

Notes: This figure compares estimated coefficients from Table 4 (Legal Abortion rates, corresponding to the third consultation when the woman attends to terminate her pregnancy) and Table 5 (Pre-Abortion visit rate, corresponding to the second consultation for mandatory counseling previous to abortion). Each point on the figure indicates the estimated coefficients accompanied by a 90% confidence interval.

4.3. Discussion

Our findings indicate that abortion legalization helped reduce adolescent fertility in Uruguay. The results are in line with those of Bailey and Lindo (2018), González et al. (2018), Mølland (2016), Pop-Eleches (2010 and 2006), and Valente (2014), who provide evidence for the US, Spain, Norway, Romania, and Nepal, respectively. These articles show that legal changes that improve access to abortion affect fertility and that the effects are more substantial among adolescents and poorer women. Because there are few places in Latin America where abortion is broadly legal, the literature is very scarce. Research for Mexico City by Clarke and Mühlrad (2016) and Gutiérrez-Vázquez and Parrado (2016) is noteworthy. Both found reductions in fertility levels, and particularly in adolescent fertility, following the expansion of access to legal abortion.¹⁷

In the case of Uruguay, the seminal study by Antón et al. (2018) found no significant effects among adolescent girls. The differing evidence with our results may be due to

¹⁷ Gutiérrez-Vázquez and Parrado (2016) split the analysis of adolescent fertility into two age groups, 14–17 and 18–19, finding significant effects only for the older group.

differences in (a) period analyzed, (b) geographical scope, and (c) data sources. First, the research by Antón et al. (2018) was very close to the reform—that is, they estimated the early effects (only considering the first two years after the law came into force). Second, they only used data for Montevideo, where pre-and post-abortion care provided in the main public maternity hospital was already operating in practice: "*The females attending this public health center, mostly with a low socioeconomic status, would have had better access to abortion than women treated at other hospitals*" (Antón et al., 2018, p. 14). Third, they used data from the Perinatal Information System (SIP, in its Spanish acronym) about the planned or unplanned nature of live births. This entails a potential problem with the filling out of medical records, because adolescent pregnancies are most often assumed to be unplanned. These reasons help to explain why the study by Antón et al. (2018) found statistically null effects for abortion legalization on adolescent fertility.

Using official data up to 2018 for the whole country, our study shows that the decriminalization of abortion in Uruguay contributed to a decline in adolescent fertility. However, different considerations need to be taken into account when interpreting our results. Among them, the importance of clandestine abortions before and after the liberalization of abortion stands out.

Before the reform, very few legal abortions were practiced, but clandestine abortion was common. According to Sanseviero et al. (2003), abortions reached 38.5 per thousand women aged 15–49 in Uruguay by 1999–2001. Hence, some of the legal abortions provided after the reform simply replaced illegal ones. This may reduce unsafe abortion-related morbidity and mortality and improve the quality of abortion records but has no direct effects on fertility rates. However, it may impact fertility, because the new law includes the supply of a contraceptive method 10 days after abortion from legal services during the post-abortion consultation (according to Ministry of Health data, 50% of women received contraception after termination).

Legal abortions are not necessarily a substitute for illegal ones, and attitudes more prone toward abortion, for example, may change in response to more liberal policy contexts. The law provided a free, legal, and safe alternative for women who cannot afford or do not want to undergo a clandestine abortion, risking their lives in unsafe settings to terminate an unwanted pregnancy. Our estimates capture both effects on adolescent fertility together—that is, the improved access to contraceptives through the

25

replacement effect of illegal abortions and the more direct fertility effects due to changes in overall abortion levels.

Another consideration is that women may still abort outside of the formal sector after the reform (as pointed out by MYSU, 2017). Our results may be biased if the number of illegal terminations is large, in particular, if clandestine and legal abortion rates are positively (negatively) related to each other, the estimated effects are likely to be biased upward (downward). It is most likely that clandestine abortions are negatively correlated with legal abortions, as states in which women face greater barriers to legal abortions foster the black market for misoprostol. Therefore, our estimates can be considered as a lower bound of the impact of abortion reform.

In line with our results, we expected a reduction in unwanted pregnancies. However, the latest available SIP data show that the share of unplanned pregnancies in adolescents' births remains stable despite the possibility of interrupting unwanted or mistimed pregnancies and the supply of free implants. We can envision at least two explanations for this paradoxical finding: (a) problems with the filling out of medical records because adolescent pregnancies are most often assumed to be unplanned; and (b) the gestational limit of 12 weeks rules out those women who recognize pregnancy later, which is common in adolescence because of psychological mechanisms of denial, difficulties in assessing future consequences, and poor sexual education (see for Uruguay Ituarte and López-Gómez, 2021; and for Mexico see Saavedra-Avendano et al., 2018).

Finally, it is important to note that abortion reform implementation met with challenges. The law requires a physician specializing in gynecology. Because most abortions are drug-induced (non-surgical), a general physician could give the prescription for misoprostol, as is the case in other countries where abortion is permitted (MYSU, 2017). One study in Mexico even shows that nurses can also provide medical abortions successfully (Díaz Olavarrieta et al., 2015). This evidence suggests that it is possible to reconcile the gynecologists' rights to exercise their conscientious objection with women's rights to have access to legal abortion services, for instance by allowing general physicians and nurses to provide medical abortions. In fact, in the case of the implant program, one of the keys to its success was that family physicians and nurse-midwives were trained in the insertion and removal of these devices, so implants were available at the primary health care level. It is worth mentioning that all reproductive-health-related tasks were solely reserved for gynecologists in the past.

5. Final remarks

This paper estimated the causal effect of abortion legalization on adolescent fertility in Uruguay, using official data on legal abortions provided after the 2012 reform. First, we employed a difference-in-differences strategy. We classified states by whether they are responsive or unresponsive to the abortion law reform using the median to obtain two groups of similar size based on the average of post-reform legal abortion rates from women aged 20–44. The results suggest that abortion reform had a negative impact on the adolescent births rates by 2.5 to 2.8 births per thousand adolescents aged 15–19. In terms of the pre-intervention average, this effect represents a 4% decrease in the adolescent fertility rate.

Then, we exploited variation in reform implementation intensity, through the estimation of fixed-effect linear regression models, regressing adolescent fertility rates on lagged legal abortion rates for the same age group. For every one-unit increase in the adolescent legal abortion rate, we estimated a reduction ranging from 0.17 to 0.24 births per 1,000 adolescents aged 15–19 years old. Because the legal abortion rate per thousand women aged 15–19 rose from zero to 11 between 2011 and 2018, our results mean a decrease of about 2.2 births per thousand girls over the whole reform period, which is quite consistent with our difference-in-differences estimates.

All models take into account the health program supporting contraceptive implants. Our estimated abortion reform effects are robust to the inclusion of measures of contraceptive implant availability. We also found that the implant program has had negative and significant effects on adolescent fertility, in line with Ceni et al. (2021).

Finally, we also examined the robustness of our findings to an alternative indicator of the reform, because our measure of abortion was based on the health facility location the woman attends to terminate her pregnancy (third consultation), but we do not have information about her state of residence. We used pre-abortion or counseling consultation (because it is more likely that a woman will attend the second consultation in her place of residence). We found similar results, addressing concerns about regional mobility across states which might have biased our estimates.

We found that abortion legalization in Uruguay caused a statistically significant reduction in birth rates for adolescent women, but it accounts for only a small proportion of the sharp decline experienced by adolescent fertility rates in recent years.

The contribution of our paper is twofold: (a) it provides rigorous evidence of the effects of abortion legalization in Uruguay exploiting official data and (b) it controls for the concurring program of contraceptive implants.

Abortion reform placed Uruguay at the forefront of the region in terms of women's rights. In addition to helping, as expected, to decrease the number of avoidable deaths and complications among women, we found that abortion legalization had unexpected positive impacts and helped to reduce adolescent fertility.

As mentioned above, Latin America and the Caribbean countries have high rates of adolescent fertility, unintended pregnancy, and an unmet need for modern contraception. The lessons from the Uruguayan case—namely, that the provision of safe abortion services coupled with expanded access to high-quality contraception may be a useful tool to tackle adolescent fertility—are relevant to many countries in the region.

6. References

Alzúa, M.L. & Velázquez, C. (2017). The effect of education on teenage fertility: causal evidence for Argentina. *IZA Journal of Migration and Development* 7: 1-23.

Amorín, D., Carril, E., & Varela-Petito, C. (2006). "Significados de maternidad y paternidad en adolescentes de estratos bajo y medios de Montevideo", A. López-Gómez (Coord.) Género y Generaciones: reproducción biológica y social de la población uruguaya, pp. 125-246. Uruguay: Trilce.

Ananat, E. O., Gruber, J., Staiger, D., & Levine, P. (2009). Abortion and selection. *Review of Economics and Statistics*, 91(1), 124-136.

Angrist, J. & Evans, W. (1999), Schooling and labor market consequences of the 1970 state abortion reforms, *Research in Labor Economics*, Emerald Group Publishing Limited, Bingley, 18: 75-113.

Angrist, J. D., & Pischke, J. S. (2008). *Mostly harmless econometrics: An empiricist's companion*. Princeton university press.

Antón, J. I., Ferre, Z., & Triunfo, P. (2018). The impact of the legalisation of abortion on birth outcomes in Uruguay. *Health economics*, 27(7), 1103-1119.

Azevedo, J,P., Favara, M., Haddock, S.E., López-Calva, L.F., Muller M., & Perova, E. (2012). *Teenage pregnancy and opportunities in Latin America and the Caribbean, On teenage fertility decisions, poverty and economic achievement.* The World Bank, Washington.

Bailey, M., & Lindo, J. (2018). Access and Use of Contraception and Its Effects on Women's Outcomes in the United States. *Oxford Handbook of Women and the Economy*: 219.

Bernstein, A., & Jones, K. (2019). *The Economic Effects of Abortion Access: A Review of the Evidence*. Institute for Women's Policy Research.

Berthelon, M. E., & Kruger, D. I. (2011). Risky behavior among youth: Incapacitation effects of school on adolescent motherhood and crime in Chile. *Journal of public economics*, 95(1-2), 41-53.

Brunet, N., Fernández-Theoduloz, G., & López-Gómez, A. (2019) Toma de decisiones y comportamientos sexuales en adolescentes. Diseño y resultados de un estudio en Uruguay (2017-2019). Montevideo: Universidad de la República, UNFPA.

Cabella, W.; Nathan, M. & Pardo, I. (2019). "La caída de la fecundidad en Uruguay entre 2015 y 2018", en: Fondo de Población de las Naciones Unidas; Ministerio de Salud Pública y Universidad de la República. *Descenso acelerado de la fecundidad en Uruguay entre 2015 y 2018. Tres estudios para su análisis*, pp. 33-70. Montevideo: UNFPA, MSP y Universidad de la República.

Cabella, W., Fernández Soto, M. Nathan, M. & Pardo, I. (2017) *Encuesta Nacional de Comportamientos Reproductivos*, 68 pp., UNFPA-INE-MIDES-PP, Universidad de la República, Montevideo.

Cabella, W. & Pardo. I. (2014) "Hacia un régimen de baja fecundidad en América Latina y el Caribe (1990-2015)" in Cavenaghi, S. & Cabella, W. (eds.) *Comportamiento reproductivo y fecundidad en América Latina: una agenda inconclusa*, ALAP Editor, Rio de Janeiro

Cardozo, S. & Iervolino, A. (2009). Adiós juventud: tendencias en las transiciones a la vida adulta en Uruguay. Revista de Ciencias Sociales, Año XXII (25): 60-81.

Casas, L., Freedman, L., Ramm, A., Correa, S., Baba, C. F., & Biggs, M. A. (2020). Chilean Medical and Midwifery Faculty's Views on Conscientious Objection for Abortion Services. *International Perspectives on Sexual and Reproductive Health*, 46(Supplement 1), 25-34.

Ceni, R., Parada, C., Perazzo, I. & Sena, E. (2021). Birth Collapse and a Large-Scale Access Intervention with Subdermal Contraceptive Implants. *Studies in Family Planning*, 52(3), 321-342.

Clarke, D., & Mühlrad, H. (2016). *The impact of abortion legalization on fertility and female empowerment: new evidence from Mexico* (No. 2016-33). Centre for the Study of African Economies, University of Oxford.

Cortés, D., Gallego, J., & Maldonado, D. (2016). On the design of educational conditional cash transfer programs and their impact on Non-education outcomes: the case of teenage pregnancy. *The BE Journal of Economic Analysis & Policy*, 16(1), 219-258.

Cortés, D., Gallego Acevedo, J., Latorre, C., Maldonado, D., Ortegón, M., & Pineros, L. (2010). *Evaluating policies to reduce teenage childbearing in Bogotá, Colombia: the effect of policies reducing costs of education faced by households.* Working Papers, PEP–PIERI.

Darney, B. G., Fuentes-Rivera, E., Polo, G., Saavedra-Avendaño, B., Alexander, L. T., & Schiavon, R. (2019). Con la ley y sin la ley/With and without the law: Utilization of abortion services and case fatality in Mexico, 2000–2016. *International Journal of Gynecology & Obstetrics*, 148(3), 369-374.

De Rosa, C.; Doyenart, M. J.; Lara, C. & Varela-Petito, C. (2016). Maternidad adolescente en barrios carenciados de Montevideo: un lugar en el mundo. *Notas de Población* 103:45-71.

Díaz Olavarrieta, C., Ganatra, B., Sorhaindo, A., Karver, T.S., Seuc, A., Villalobos, A., García, S.G., Pérez, M., Bousieguez, M., & Sanhueza, P. (2015). Nurse versus physician-provision of early medical abortion in Mexico: a randomized controlled non-inferiority trial. *Bulletin of the World Health Organization*, 93 (4), 249 - 258.

Donohue III, J. J., & Levitt, S. D. (2001). The impact of legalized abortion on crime. *Quarterly Journal of Economics*, 116(2), 379–420.

Driscoll, J. C., & Kraay, A. C. (1998). Consistent covariance matrix estimation with spatially dependent panel data. *Review of economics and statistics*, 80(4), 549-560.

ECLAC (2011) Panorama social de América Latina 2011, Economic Commission for Latin America and the Caribbean, Santiago de Chile.

Fiol, V., Briozzo, L., Labandera, A., Recchi, V. & Piñeyro, M. (2012). Improving care of women at risk of unsafe abortion: Implementing a risk-reduction model at the Uruguayan-Brazilian border. *International Journal of Gynecology & Obstetrics*. 118: S21-S27.

Fiol, V., Rieppi, L., Aguirre, R., Nozar, M., Gorgoroso, M., Coppola, F., & Briozzo, L. (2016). The role of medical abortion in the implementation of the law on voluntary termination of pregnancy in Uruguay. *International Journal of Gynecology & Obstetrics*, *134*(S1), S12-S15.

Fink, L. R., Stanhope, K. K., Rochat, R. W., & Bernal, O. A. (2016). "The fetus is my patient, too": attitudes toward abortion and referral among physician conscientious objectors in Bogotá, Colombia. *International perspectives on sexual and reproductive health*, 42(2), 71-80.

Foster, D. G. (2019). Introduction to the Turnaway Study. *Advancing New Standards in Reproductive Health, University of California, San Francisco, Issue Brief.*

González, L., Jiménez-Martín, S., Nollenberger, N., & Vall-Castello, J. (2018). *The effect of abortion legalization on fertility, marriage and long-term outcomes for women.* Universitat Pompeu Fabra, Barcelona, Economics Working Paper Series, Working Paper No. 1607.

Gruber, J., Levine, P., & Staiger, D. (1999). Abortion legalization and child living circumstances: who is the "marginal child"?. *The Quarterly Journal of Economics*, 114(1), 263-291.

Gutiérrez-Vázquez, E.Y. & Parrado, E.A. (2016), Abortion Legalization and Childbearing in Mexico. *Studies in Family Planning*, 47: 113-128.

Hoechle, D. (2007). Robust standard errors for panel regressions with cross-sectional dependence. *The stata journal*, 7(3), 281-312.

Ituarte, M. L., & López-Gómez, A. (2021). Las adolescentes frente a la decisión de interrumpir un embarazo en un contexto de aborto legal. *Cadernos de Saúde Pública*, 37, e00235219.

Labandera, A., Gorgoroso, M., & Briozzo, L. (2016). Implementation of the risk and harm reduction strategy against unsafe abortion in Uruguay: From a university hospital to the entire country. *International Journal of Gynecology & Obstetrics*, 134(S1), S7-S11.

López-Gómez, A.; Brunet, M.; & Couto, M. (2019). Estudio sobre descenso de la fecundidad y factores explicativos en Uruguay 2015-2017: políticas públicas implementadas en el período analizado. In UNFPA, MSP & UDELAR (eds.). *Descenso acelerado de la fecundidad en Uruguay entre 2015 y 2018. Tres estudios para su análisis.* Montevideo: UNFPA, MSP & UDELAR.

López-Gómez, A., & Couto, M. (2017). Profesionales de la salud, resistencia y el cambio en la atención a mujeres que deciden abortar en Uruguay. *Descentrada*, *1*(2), e021-e021.

López-Gómez, A. (2015). Profesionales de la salud y aborto. Posibilidades y resistencias a la incorporación del enfoque de derechos en las prácticas de las/os profesionales de la salud. In Ramos, S. (comp.). *Investigación sobre aborto en América Latina y el Caribe. Una agenda renovada para informar políticas públicas e incidencia.* Buenos Aires: CEDES, 19-40.

López-Gómez, A. (2014). Aborto legal en Uruguay: el largo proceso para garantizar el acceso a la salud y los derechos reproductivos de las mujeres, chapter 6. In Cavenaghi, S. & Cabella, W. (eds.) *Comportamiento reproductivo y fecundidad en América Latina: una agenda inconclusa*. ALAP, Río de Janeiro.

López-Gómez, A., & Abracinskas, L. (2009). *El debate social y político sobre la ley de defensa del derecho a la salud sexual y reproductiva.* Cuadernos del UNFPA, serie Divulgación.

Mautone, M., & Rodríguez-Almada, H. (2013). Objeción de conciencia en el ámbito de la salud. *Revista Médica del Uruguay*, 29(1), 40-42.

Mølland, E. (2016). Benefits from delay? The effect of abortion availability on young women and their children. *Labour Economics*, 43, 6-28.

MYSU (2017). Estado de situación de los servicios de salud sexual y reproductiva y aborto legal en 10 de los 19 departamentos del país. Sistematización de resultados. Estudios Observatorio MYSU 2013-2017. Montevideo, MYSU.

MYSU (2016a). Informe sobre el estado de situación de los servicios de salud sexual y reproductiva y aborto en las instituciones de salud de los departamentos de Florida, Maldonado y Rivera. Observatorio Nacional en Género y Salud Sexual y Reproductiva de Mujer y Salud en Uruguay. Montevideo, MYSU.

MYSU (2016b). Informe sobre estado de situación de los servicios de salud sexual y reproductiva y aborto en las instituciones de salud de los departamentos de Montevideo, Rocha y Cerro Largo. Observatorio Nacional en Género y Salud Sexual y Reproductiva de Mujer y Salud en Uruguay. Montevideo, MYSU.

MYSU (2015). Informe sobre el estado de situación y desafíos en salud sexual y reproductiva y aborto en los departamentos de Paysandú, Río Negro y Soriano. Observatorio Nacional en Género y Salud Sexual y Reproductiva de Mujer y Salud en Uruguay. Montevideo, MYSU.

MYSU (2014). Estado de situación y desafíos en aborto y salud sexual y reproductiva en el departamento de Salto. Observatorio Nacional en Género y Salud Sexual y Reproductiva de Mujer y Salud en Uruguay. Estudio 2013-2014: Montevideo, MYSU.

Nathan, M., Pardo, I. & Cabella, W. (2016). Diverging patterns of fertility decline in Uruguay. *Demographic Research*, 34(20): 563-586.

Nathan, M. (2015). La lenta transición hacia un régimen de fecundidad tardía en Uruguay: los cambios en la edad al primer hijo entre 1978 y 2011. *Revista Latinoamericana de Población*, (17), 37-60.

Novella, R., & Ripani, L. (2016). Are you (not) expecting? The unforeseen benefits of job training on teenage pregnancy. *IZA Journal of Labor & Development*, 5(1), 1-18.

Ortiz-Millán, G. (2018). Abortion and conscientious objection: rethinking conflicting rights in the Mexican context. *Global Bioethics*, 29(1), 1-15.

Pop-Eleches, C. (2010). The supply of birth control methods, education, and fertility evidence from Romania. *Journal of Human Resources*, 45(4), 971-997.

Pop-Eleches, C. (2006). The impact of an abortion ban on socioeconomic outcomes of children: Evidence from Romania. *Journal of Political Economy*, 114(4), 744-773.

Remez, L., Mayall, K., & Singh, S. (2020). Global Developments in Laws on Induced Abortion: 2008–2019. *International Perspectives on Sexual and Reproductive Health*, 46(Supplement 1), 53-65.

Ribar, D. C. (1994). Teenage fertility and high school completion. *The Review of Economics and Statistics*, 413-424.

Rodríguez-Vignoli, J. (2017) Fecundidad no deseada entre las adolescentes latinoamericanas Un aumento que desafía la salud sexual y reproductiva y el ejercicio de derechos. Serie Población y Desarrollo, CEPAL-UNFPA, Santiago de Chile.

Rodríguez-Vignoli, J., & Cavenaghi, S. (2014). Adolescent and youth fertility and social inequality in Latin America and the Caribbean: what role has education played. *Genus*, 70(1), 1-25.

Saavedra-Avendano, B., Schiavon, R., Sanhueza, P., Rios-Polanco, R., Garcia-Martinez, L., & Darney, B. G. (2018). Who presents past the gestational age limit for first trimester abortion in the public sector in Mexico City?. *PLoS One*, 13(2), e0192547.

Sanseviero, R., Rostagnol, S., Guchin, M., & Migliónico, A. (2003). *Condena, tolerancia y negación: el aborto en Uruguay*. Universidad para La Paz, Montevideo, UY.

Singh, S., Remez, L., Sedgh, G., Kwok, L., & Onda, T. (2018). Abortion worldwide 2017: uneven progress and unequal access. New York: Guttmacher Institute; 2018.

United Nations (2019). World Population Prospects 2019: Online Edition. Rev.1, Department of Economic and Social Affairs, Population Division.

Valente, C. (2014). Access to abortion, investments in neonatal health, and sex-selection: Evidence from Nepal. *Journal of Development Economics*, 107, 225-43.

Varela-Petito, C., Pardo, I., Lara, C., Nathan, M., & Tenenbaum, M. (2014). La fecundidad en el Uruguay (1996-2011), chapter 3. In Calvo, J. J. (ed.) *Atlas Sociodemográfico y de la Desigualdad del Uruguay*, Ediciones Trilce, Montevideo.

Varela-Petito, C., Fostik, A. & Fernández, M. (2012). Maternidad en la juventud y desigualdad social. Cuadernos del UNFPA, Año 6, Nº 6. Montevideo: UNFPA.

Videgain, A. K. (2006). Análisis de los cambios en la transición a la adultez en mujeres de distintas cohortes en contexto de cambios sociales en el Uruguay contemporáneo. Tesis de Maestría en Demografía. México D.F.: El Colegio de México.

Zamberlin, N., Romero, M., & Ramos, S. (2012). Latin American women's experiences with medical abortion in settings where abortion is legally restricted. *Reproductive health*, 9(1), 34.

Appendix. Conscientious objection

Conscientious objection has been identified as an important barrier to abortion access in many countries where abortion is legally permitted under broad criteria. The situation in Latin America and the Caribbean is worrying, where nearly every country has highly restrictive abortion laws (Remez et al., 2020; Singh et al., 2018).¹⁸ Even where abortion has been decriminalized, conscientious objection is one of the strongest barriers to its implementation (for Mexico City, see Ortiz-Millán, 2018). In Uruguay, this issue was identified as the main barrier to access to legal abortion.

There is no complete information at the state level about the evolution of conscientious objectors. The little official information available shows that, in the first year after the reform (2013), 30% of Uruguayan gynecologists declared themselves to be objectors. This rose to 40% in September 2015 (data provided by the Ministry of Health).

Data collected by the nongovernmental organization "Women and Health Uruguay" (*MYSU* in its Spanish acronym) regarding conscientious objectors show a large disparity and alarming figures in some states. So far, MYSU has collected data at different points in time in about 10 of the 19 states, covering 64% of the national population. Figure 6 shows the distribution of conscientious objectors across the country.

After legalization, all 12 gynecologists residing in the state of Salto claimed conscientious objection, leaving women in Salto without access to abortion services until September 2013, when the Ministry of Health began to send a non-objecting gynecologist for the third consultation once a week (the ratio of objectors was reduced to 92%). Currently, 80% of gynecologists are conscientious objectors in Salto. There are also worrying figures in the states of Paysandú (87%); Soriano (82%, rising to 100% in the principal town, Mercedes); Rivera (above 60%),; Río Negro (43%, rising to 100% in the second most important city, Young); and Cerro Largo (30%–60%, but in Río Branco city the only non-objecting gynecologist is not on call every day of the month). In the state of Rocha, the ratio of objectors is below 30%, but it rises to 100% in the third most important city, Castillos. Finally, in the states of Montevideo (the

¹⁸ López-Gómez (2015) provides a literature review on attitudes toward abortion and conscientious objection among health professionals in several Latin American countries: Brazil, Mexico, Argentina, Paraguay, Jamaica, and Uruguay. See Fink et al. (2016) for Colombia and Casas et al. (2020) for Chile.

country's capital), Maldonado, and Florida, fewer than 30% of gynecologists declare conscientious objection (MYSU, 2014, 2015, 2016a, 2016b, 2017).

In those cities or towns where 100% of gynecologists invoke conscientious objection, women must be referred to other cities of the country for treatment. From the city of Young, they are referred to Fray Bentos (the state capital, about 100 km away); from Mercedes to other cities in the state, such as Dolores or Cardona (39 km and 96 km away, respectively) and also to Fray Bentos (at 35 km); and from Castillos to the city of Rocha (the state capital, 57 km away). On those days when the gynecologist is not available or on call in the city of Río Branco, women are referred to Melo (the state capital, 80 km away). In all cases, when the services are overcrowded, women are referred to the Pereira Rossell Hospital in Montevideo. Their travel costs must be covered by the institution that failed to provide the service, although this does not always occur (MYSU, 2014, 2015, 2016a, 2016b, 2017).



Figure 6. Distribution of conscientious objectors (COs) across the country

Note: So far, MYSU has collected data at different points in time in about 10 of the 19 states, covering 64% of the national population. Source: Own elaboration based on MYSU (2014, 2015, 2016a, 2016b, 2017).

In addition to the state-level variation, there is also significant variation over time. Law 18987 on Voluntary Termination of Pregnancy recognizes health professionals' right to

refrain from performing abortions. However, the detailed rules on the procedures to be followed when seeking recognition as a conscientious objector have faced stiff resistance and have even been changed several times. Figure 7 summarizes the legislation changes.

Regulatory Decree 375/012 established that gynecologists must inform the authorities of all health facilities in which they work of their conscientious objection in writing and that it applies only to the third of the four consultations required. This means that gynecologists with a pro-life stance may assist women before and after pregnancy termination, placing them in a vulnerable situation. This is of particular concern during the counseling consultation (López-Gómez and Couto, 2017).

In June 2013, the pro-life movement organized a non-binding referendum, but the threshold required for repealing Law 18987 on Voluntary Termination of Pregnancy was not reached. One month after the failure of this referendum, a group of 20 gynecologists¹⁹ (including two of the three titular professors of gynecology at the main public university) submitted a request for partial annulment of the regulatory decree to the administrative court (*TCA* to use its Spanish acronym). Their main argument was that the decree impeded the exercising of the right of conscientious objection enshrined in Law 18987 and in the National Constitution.²⁰

In August 2014, those articles in the regulatory decree regarding conscientious objection were suspended by the TCA until a final decision was made. Health professionals were released from caring for women before and after abortion. The requirement to provide written notification of conscientious objection was also suspended. In August 2015, the TCA's final decision extended the right of conscientious objection to paraclinical and administrative staff, who were not previously included. It also extended the right to pre- and post-abortion consultations. However, the obligation to report conscience-based refusal to provide abortions in writing was reinstated.

¹⁹ Our empirical analysis could benefit from information on this group of gynecologists, particularly their place of residence. Unfortunately, we do not have such information.

²⁰ It is important in this matter to draw a distinction between conscientious objection and civil disobedience: "Conscientious objection (...) is an individual act that does not aim to change the law, and neither does it seek for others to revise their decision. It is different from civil disobedience since it consists of a group of people who state they do not want to observe a legal provision, it aims to abolish it and to exert a group pressure on the government" (Mautone and Rodríguez-Almada, 2013).

In April 2016, the Ministry of Health passed two new ordinances introducing further changes. Those ministerial ordinances guarantee the confidentiality of the names of conscientious objectors. In addition, they require that, during the second of the four consultations, the three members of the board must provide information in one single meeting. In other words, the legal abortion services need to coordinate the schedules of a non-objecting gynecologist, a social worker, and a psychologist. This presents great difficulty in those cities or towns that are facing a shortage of professionals.

In sum, the scant information available suggests that the share of Uruguayan gynecologists claiming conscientious objection is substantial enough to be a meaningful barrier to safe, legal abortion access in many states. This problem was exacerbated during those periods marked by the administrative court (TCA) decisions, when the use of conscientious objection was subject to lax or less strict criteria.

Figure 7. Legislation changes

